Author's response to reviews

Title: Net benefits: assessing the effectiveness of clinical networks in Australia through qualitative methods

Authors:

Frances C Cunningham (f.cunningham@unsw.edu.au)
Geetha Ranmuthugala (g.ranmuthugala@nsw.edu.au)
Johanna I Westbrook (j.westbrook@unsw.edu.au)
Jeffrey Braithwaite (j.braithwaite@unsw.edu.au)

Version: 6 Date: 13 September 2012

Author's response to reviews:

Dear Editor,

I have amended a typographical error on p. 14 of the manuscript, in the second sentence of the Results section. I have deleted 'identified by'.

Yours sincerely

Dr Frances Cunningham
Senior Research Fellow
Centre for Clinical Governance Research
Australian Institute of Health Innovation
University of New South Wales
Sydney NSW 2052
Australia